ELSEVIER

Contents lists available at ScienceDirect

Ecosystem Services



journal homepage: www.elsevier.com/locate/ecoser

Full Length Article

Measuring non-use values to proxy conservation preferences and policy impacts

Marco Nilgen^{a,*}, Julian Rode^b, Tobias Vorlaufer^c, Björn Vollan^d

^a University of Marburg, Department of Economics, Research Group Sustainable Use of Natural Resources, Am Plan 1, 35032 Marburg, Germany

^b Helmholtz-Centre for Environmental Research, Department of Environmental Politics, Leipzig, Germany

^c Leibniz Centre for Agricultural Landscape Research (ZALF), Working Group Governance of Ecosystem Services, Germany

^d University of Marburg, Department of Economics, Research Group Sustainable Use of Natural Resources, Am Plan 1, 35032 Marburg, Germany

ARTICLE INFO

Keywords: Conservation preferences Conservation policy Donations Contingent valuation Deliberative monetary valuation Hypothetical bias

ABSTRACT

The behavior of local natural resource users is not only affected by economic incentives but also by a diverse set of motivations and underlying values. These non-monetary drivers are crucial in safeguarding long-term positive conservation outcomes. However, measuring these factors still constitutes a significant challenge. Building on lessons learned from established methodology such as attitudinal or behavioral measures, we showcase how a contingent valuation method and experimental donation tasks can be used to measure relative changes in nonuse values and are a good proxy for conservation preferences. We exemplify this approach within the context of a case study in northern Namibia, where it was employed to investigate whether exposure to a community-based conservation program affects individual conservation preferences. Our findings show that our approach can serve as a complement to established measures for conservation preferences while avoiding some of the existing pitfalls such as demand effects or costly data collection associated with behavioral and attitudinal measures.

1. Introduction

Environmental awareness or inclination toward conservation has a history of being measured via agreement to attitudinal or value-based statements (Nilsson et al., 2020). These approaches have faced criticism, primarily because of growing concerns surrounding phenomena like the value-action gap (Kollmuss & Agyeman, 2002) as well as the reliability and validity of self-reported measures, especially in contexts where respondents face complex trade-offs between environmental protection and economic goals (Kenny, 2021). Researchers have also raised doubts about the accuracy of responses provided by survey participants, as they may align their answers with the perceived social desirability (Milfont, 2009; Vesely & Klöckner, 2020). Likewise, measures of actual conservation behavior are often confounded with incentives, rules, and regulations that are already in place. Thus, people might carry out a desired behavior, but not due to their 'conservation preferences'. In this article, we build on terminology and empirical traditions from economics to showcase an application of preference measurement that allows for circumventing some of the challenges associated with attitudinal or behavioral measures of conservation preferences. We highlight how the contingent valuation method, when focusing on non-use values, can add insights into whether and how a certain policy has influenced the way people relate to conserving ecosystems. Our case study from community-based natural resource management (CBNRM) communities in northern Namibia and neighboring villages that are not part of the policy provides evidence that conservation preferences have not increased despite the conservation policy being in place for more than ten years.

What are conservation preferences? Economists conceptualize preferences as the drivers of individual decision-making and behavior (Falk et al., 2018; D. M. Hausman, 2011). If a person has positive conservation preferences, this person can thus be expected to be supportive of conservation interventions and also to individually act in favor of conservation outcomes (Selinske et al., 2020). E.g., in a choice situation characterized by clear individual trade-offs between environmental and economic outcomes, a person with stronger conservation preferences would be willing to accept more economic losses in order to ensure a positive conservation outcome than a person with lower conservation preferences. As we will further clarify, conservation preferences should be clearly distinguished from preferences for personal economic benefits

* Corresponding author.

https://doi.org/10.1016/j.ecoser.2024.101621

Received 4 July 2023; Received in revised form 29 March 2024; Accepted 8 April 2024

E-mail addresses: marco.nilgen@wiwi.uni-marburg.de (M. Nilgen), julian.rode@ufz.de (J. Rode), tobias.vorlaufer@zalf.de (T. Vorlaufer), bjoern.vollan@wiwi. uni-marburg.de (B. Vollan).

^{2212-0416/© 2024} The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

people may derive or expect to receive from conservation activities. Otherwise, they might reflect a wide array of values that people attribute to nature, which have been conceptualized in various ways (e.g., intrinsic, instrumental, and relational values) (Chan et al., 2016; Pascual et al., 2017, 2023). Within the Total Economic Value framework of nature valuation (Anderson et al., 2022, Fig. 2.13) conservation preferences could thus encompass all types of non-use values as well as direct and indirect use values from natural processes that do not lead to a direct personal economic benefit (such as selling fish or fruits on the market or generating income through eco-tourism). The academic discourse on the potential relevance of non-use values in conservation policymaking, as well as how and to which degree they should be incorporated into economic cost-benefit analyses dates back to the 1960s (Krutilla, 1967; Weisbrod, 1964). Following economic theory, an insufficient acknowledgment of non-use values in decision-making governing conservation policies would lead to an under-provision of the natural resources providing them. While discussions about problems inherent to the exact measurement of non-use values remained a point of contention over the years (see e.g., Carson, 2012; Hausman, 2012), they continue to be an important concept in climate policy on a global scale and their full incorporation into related cost-benefit calculations would have significant policy impacts (Bastien-Olvera & Moore, 2021).

Are conservation preferences important to consider when implementing, managing, and evaluating conservation policies? Conservation practitioners would likely agree that the level of support for conservation actions and the likelihood to act in line with conservation goals are desirable outcomes of many policy interventions. Especially in cases where economic benefits to local communities are only temporary (e.g., via donor support in implementation stages), or fail to materialize at all, an increase of conservation preferences induced by these policies is thus crucial for achieving longer-term positive conservation outcomes (Hayes et al., 2022).¹ Particularly interesting applications of conservation preferences as policy impacts can be found within the context of community-based conservation programs such as the above-mentioned CBNRM or Payments for Ecosystem Services (PES). Policy approaches like this are frequently presented as win-win solutions, in which both biodiversity conservation and economic well-being or development can be promoted simultaneously. However, scenarios that effectively overcome the complex tradeoffs between those two objectives may be relatively rare and difficult to realize in the first place (Hegwood et al., 2022; McShane et al., 2011; Muradian et al., 2013). Table 1 illustrates and describes four theoretical cases, reflecting outcomes of conservation behavior in response to conservation initiatives or policies, that differ both in the realization of economic benefits for the local population and their influence on conservation preferences. Case C represents the case in which CBNRM programs include a bundle of interventions such as awareness raising and information campaigns that strengthen conservation preferences irrespective of any economic benefits (Green et al., 2019). Cases A and B represent incentive-based policies that provide economic benefits to the community. The effect of such policies on conservation preferences can go both ways depending on their design and the socio-cultural context - incentives have been shown to

Table 1

Dependency of Conservation Behavior on Conservation-related Preferences and Economic Benefits. In the presence of economic net benefits (cases A and B), positive conservation actions do not depend on the presence of conservation preferences. However, one would expect communities with strengthened conservation preferences (case A) to be more receptive toward additional conservation actions that do not generate economic returns per se. A decrease in conservation preferences may even result in a net negative effect on conservation behavior (case B). When conservation actions are conditional on enhanced conservation preferences (case C). Finally, one would not expect positive conservation behavior, if no financial benefits are generated nor conservation preferences are enhanced (case D).

Net effect on conservation behavior		Increase in conservation preferences of resource users (e.g. internalization of ecosystem benefits)	
		Yes	No / Unclear
Economic net benefits to resource users (e.g. payments, labor)	Yes	A (++)	B (+, 0, or -)
	No	C (+)	D (0 or –)

potentially enhance (crowd-in) or decrease (crowd-out) non-economic motivations for conservation (Blanco et al., 2023; Ezzine-de-Blas et al., 2019; Rode et al., 2015).

In this paper, we conceptualize and measure conservation preferences within a case study on community conservancies in Namibia, one of the most prominent and long-standing CBNRM programs in Sub-Saharan Africa (Boudreaux & Nelson, 2011). Our study investigates potential changes to conservation preferences in a scenario in which the aforementioned economic incentives have widely not materialized, thereby increasing the importance of considering conservation preferences (case C or D).

In our case study, we employed a contingent valuation scenario for a one-time donation to a Namibian conservation project, concerned with monitoring and saving remaining populations of the lappet-faced vulture (Torgos Tracheliotos), an endangered bird species native to almost all of Namibia. We specifically chose this scenario and designed it in such a way that it mostly captures non-use values or indirect use values at most. Within the Total Economic Value framework of nature valuation, non-use values are distinguished from use-values in that they provide no direct benefits to the individual in question. Non-use values are typically further differentiated between those that origin from the satisfaction of knowing that the good in question will be available to others now (altruistic values) or to future generations (bequest values). A third group of non-use values are so-called existence values, which are solely derived from the satisfaction of knowing that an environmental good or natural resource exists (Manero et al., 2022, Fig. 1.; Otrachshenko et al., 2022). While indirect use values are typically listed as a use-value category, they provide no directly perceivable benefits to the individual (e.g., in the form of tradeable goods or recreational activities) but provide important ecosystem functions (Pascual et al., 2010). However, in our case, the specific vulture species is very rarely found in the study region and therefore the role of indirect use values is largely negligible. We thus argue that the amount donated to the vulture conservation cause can be regarded as a valid measure of conservation preferences. This is because the trade-off inherent to the donation decision (giving up the potential utility derived from spending the money for the benefit of nature) is not confounded with any direct benefit from the donation through use values of vultures. Keeping in mind that conservation preferences for all nature within the community land may involve many more value components associated with species and ecosystems, the absolute value of conservation preferences for the particular vulture species can certainly not be equated with the absolute value

¹ Conservation preferences have, to our knowledge, not yet been treated explicitly as policy impacts in economics. One reason for this could be that economics traditionally conceptualized individual preferences to be stable over time, similarly to how personality traits are viewed in psychology. By definition, this would not make them subject of policy evaluations. Individual attitudes, on the other hand, are considered to be less stable (Achen, 1975) and to be changing with experience or depending on the specific context (Gifford et al., 2011). The assumption of stable preferences has always been contested within the social sciences and preference instability has been captured within concepts such as 'endogenous' (Bowles, 1998) or 'adaptive' preferences (von Weizsäcker, 2005). Once preferences are no longer regarded as a stable character trait, they can indeed be considered as outcome variables of an (environmental) policy process.

of conservation preferences for nature. Yet, we argue that it can serve as a conservative and clean indicator for measuring changes in conservation preferences over time and across study sites.

The case study further enables us to scrutinize two methodological key issues related to economic measures of conservation preferences: First, observing both hypothetical willingness to pay (WTP) statements and real donations on an individual level allows us to evaluate potential differences in the hypothetical bias between study sites. Second, we use our case study to experimentally investigate the effect of group deliberation on the measures of conservation preferences.

The remainder of the paper is structured in the following way: Section 2 discusses the concept of conservation preferences, and also provides a critical evaluation of different types of instruments for their measurement. Section 3 presents an application of our proposed measurement approach involving communal conservancies in the Kavango-East Region, Namibia. The section is further subdivided into subsections describing the study context, the design of the contingent valuation scenario as well as an analysis section describing our hypotheses, econometric approach and the study results. Section 4 discusses the case study findings and Section 5 concludes.

2. Measuring conservation preferences and related concepts

Conservation preferences share conceptual similarities to key concepts from psychology such as 'motivations' (Ryan & Deci, 2000), values (Shiell et al., 1997), and 'attitudes' (Nilsson et al., 2020) for conservation. In this paper, we cannot fully alleviate the complexity of delineating these different concepts. Instead, we intend to clarify in particular the role of economic methods among the plurality of related tools from different disciplines to measure conservation preferences. The following paragraphs explain the different approaches for measuring conservation preferences.

2.1. Attitudinal measures

Psychologists traditionally investigate how values, attitudes, and norms change, which are assumed to influence behavioral intentions and actual behavior (de Groot & Steg, 2008; Fishbein & Ajzen, 1975; Stern & Dietz, 1994). In particular, attitudinal measures are a common tool in psychological research on biology conservation and are used to evaluate the effectiveness of conservation programs (Nilsson et al., 2020). Attitudinal statements rely on Likert-scale ratings, using items such as the New Ecological Paradigm (NEP) scale (Dunlap et al., 2000) or via items directly relevant to the specific conservation case, as e.g., individual attitudes on (un)desired conservation behavior, endangered species or wildlife management practices (Nilsson et al., 2020).

The main virtue of eliciting attitudes to evaluate conservation preferences is that it is quick, cheap, easily implementable, and, depending on the use of standardized scales, also comparable across different study contexts. Widely applied scales, such as e.g., the NEP, have been typically tested for reliability and validity, however often only in specific cultural (i.e., western) contexts. While attitudes have been shown to predict pro-environmental behavior, on average they only do so imperfectly (Armitage & Conner, 2001); the phenomenon commonly referred to as "attitude-behavior-gap" in environmental economics and psychological literature (Farjam et al., 2019). Experimental economists have also pointed out that attitudinal scales are also especially prone to demand effects (Zizzo, 2010). In this case, respondents would consciously or unconsciously provide answers in line with their conceptions about the researcher's expectations. For example, respondents may provide systemically biased answers if they expect that these answers increase the chances of attracting or retaining conservation projects and associated funding.

2.2. Behavioral intentions

As a way to get closer to conservation behavior, many studies have used self-reported behavior (e.g., "Do you typically engage in wildfire control measures?") or behavioral intentions (e.g., "In the coming month, do you plan to engage more regularly in wildfire control measures?") to approximate people's conservation preferences. Survey items of this type can be regarded as advantageous over scale-based attitudinal statements as they can be adjusted to the specific context of the conservation intervention in question. However, as purely hypothetical statements, they suffer from similar drawbacks as the attitudinal measures, namely being prone to demand effects and an unclear correspondence to real behavior (Webb & Sheeran, 2006). These issues may be particularly pronounced when the behavioral intentions relate to potentially incriminating practices such as poaching or illegal logging. Additionally, in contrast to purely attitudinal statements, behavioral intentions can be already influenced by economic constraints, i.e., material pay-offs associated with specific behavior. Conservation programs may alter these parameters and thus affect behavioral intentions without changing underlying conservation preferences.

2.3. Stated preferences: Willingness-to-pay (WTP)

Economists have a long tradition of measuring preferences based on monetary values, either by observing actual behavior (e.g., purchases on markets) or by eliciting stated preferences, e.g. in the form of willingness-to-pay (WTP). WTP is commonly understood as the maximum monetary amount an individual is willing to forego in order to e.g., buy a product or service. These approaches are thought to be superior as respondents face a trade-off between conservation and something else. The measurement of conservation preferences should not automatically assume win–win constellations between environmental and economic outcomes. Instead, it should acknowledge that the conservation of natural resources, at least initially, can in fact be detrimental to economic goals. Attitudinal measures or behavioral intentions neglect such trade-offs or at least have difficulties in disentangling individual motives (Kenny, 2021), while economic approaches enable researchers to take them into account.

While the discrete choice experiment is nowadays the more popular method to elicit WTP estimates for natural resources (Chaikaew et al., 2017; Dias & Belcher, 2015), the 'contingent valuation' approach (Carson, 2012b; J. Hausman, 2012b; Kling et al., 2012), where study participants are directly asked to state their hypothetical WTP for an environmental good, remains in use as well (e.g. Otrachshenko et al., 2022). While in the most typical case, this approach is used to derive an economic value of environmental goods (Carson et al., 2003; Otrachshenko et al., 2022) to inform environmental policy, individual WTP can also be regarded as a proxy for individuals' conservation preferences. In the latter case, the valuation scenario should be carefully designed. For example, the WTP question could be framed as a one-off payment to promote the conservation of a nature reserve or an endangered species. Importantly, all respondents (both targeted and not targeted by the policy or intervention at hand) should face similar economic (dis)incentives towards the respective conservation good. Rather than focusing on the absolute monetary levels WTP - in this application - is a relative, comparative measure for conservation preferences between populations (e.g., targeted and not targeted by a conservation project or policy) or over time.

Many economists have been wary of mere statements of preferences, which they consider "cheap talk" (Farrell and Rabin, 1996), and have favored 'revealed preferences' measures based on real-world observations of actual behavior (Frey and Stutzer, 1999). Nevertheless, another line of economics research, especially related to the valuation of environmental goods, relies on contingent valuation methods, where individuals' hypothetical "willingness-to-pay" statements indicate their 'stated preferences' (Carson, 2012; Hausman, 2012). The fact that such WTP statements are purely hypothetical renders them conceptually similar to the previously discussed behavioral intentions - a person states the intention to act in a certain way (making a payment in this case) - and they, therefore, are prone to similar problems. The contingent valuation method, as well as the closely related discrete choice experiment methodology are under constant criticism for its susceptibility to a phenomenon frequently referred to as 'hypothetical bias', i.e., that studies applying such methods would systematically lead to the overestimation of environmental values (J. Hausman, 2012b; Hensher, 2010). A number of meta-analyses from recent years investigate this phenomenon and provide evidence that the magnitude of hypothetical bias can at least be mitigated to some degree by the use of so-called cheap talk scripts or consequentiality scripts, where survey respondents are informed about the hypothetical bias or consequentiality of WTP bids before the value elicitation (Andor et al., 2017; Foster & Burrows, 2017; J. Penn & Hu, 2019; J. M. Penn & Hu, 2018). Other means proposed in the literature to potentially tackle such biases are to ensure task familiarity among respondents applied (Schläpfer and Fischhoff, 2012), using oath scripts prior to the value elicitation (Stevens et al., 2013), or controlling for certain respondent characteristics in the analysis of hypothetical WTP studies (Wuepper et al., 2019).

Other criticisms of the contingent valuation method revolve around the insensitivity to the scope of WTP (Lopes & Kipperberg, 2020), that respondents oftentimes do not have well-defined preferences for goods that are not commonly exchanged in market settings (Hsee & Rottenstreich, 2004; Spash, 2007), or that they might even protest against monetary attribution to nature goods altogether (Spash et al., 2009). Financial constraints may also render money a questionable unit for conservation preferences, e.g., when poor communities do not have the ability to support conservation activities financially (Martínez-Alier, 2002). Thus, absolute levels may be understated and income differences between study sites may render comparisons of hypothetical WTP measures in absolute terms difficult. However, in contrast to the other approaches, WTP statements allow for a relative comparison of conservation preferences between different study sites, while clearly reflecting the trade-offs between financial and conservation outcomes inherent to conservation. The ease of comparability in combination with the high degree of flexibility in designing WTP scenarios enables researchers to easily enhance this approach with additional experimental interventions, which can yield new insights into how and what might drive changes in conservation preferences.

2.4. Behavioral measures (revealed preferences)

Intuitively one may presume that the evolution of conservation preferences over time would ideally be monitored by observing realworld conservation behavior and assessing the direct links to conservation outcomes, such as a reduction of deforestation or poaching within a randomly assigned trial. Monitoring such outcomes, however, is in many cases practically infeasible, or difficult and costly to implement (Nilsson et al., 2020). While preference measures based on observed behavior provide more objective and reliable behavioral data than the aforementioned approaches based on self-reported statements and intentions, they come with downsides. They are relatively costly to implement, especially when conservation behavior is supposed to be monitored over a longer period of time (Floress et al., 2018). Moreover, conservation projects commonly aim to alter the costs and benefits associated with conservation actions. Differences in conservation behavior between intervention and control populations are likely to reflect changes in monetary (dis)incentives, regulations, or social norms rather than a fundamental shift in the underlying conservation preferences. In other cases, conservation interventions may have already been proven successful in improving environmental quality to an extent that additional behavioral adjustments can hardly be expected, even though the intervention might have in principle strengthened conservation preferences.

2.5. Donations

Real monetary donations towards a conservation cause conceptually represent a specific behavioral measure. Such measures are typically assumed to measure altruism coming at a real personal cost and they have been used in multiple studies from the field of experimental economics to approximate altruistic preferences towards social or environmental causes (e.g. Blanco et al., 2012; Champ et al., 1997; Rode et al., 2008). The cause for which people are asked to donate can be selected in a targeted way such that donations capture underlying conservation preferences as opposed to economic dis(incentives) from conservation. As for WTP measures above, financial constraints may however render money a questionable measure for conservation preferences, e.g., when poor communities do not have the ability to support conservation activities financially (Martínez-Alier, 2002). Recently, studies have employed donation tasks in combination with contingent valuation scenarios in experiments to investigate how WTP bids to support public causes are driven by behavioral factors (Bouma & Koetse, 2019). In the remainder of this article, we aim to showcase how a combination of both WTP measures and donations can be applied to measure and investigate underlying conservation preferences based on a case study on CBNRM in Namibia (Table 2).

3. Case study

3.1. Context of CBNRM in Namibia

CBNRM in Namibia is often considered a success story of nature conservation in Africa (a detailed description of CBNRM in Namibia and our selected conservancies can be found in the Supplementary Online Material (SOM), Section 1). Conservancies within Namibia's national CBNRM program cover a total area of 166,045 km², amounting to 52.9% of all communal land in Namibia. However, although some conservancies have successfully developed income sources from tourism or hunting, substantial economic benefits have not materialized for the majority of conservancies. As of 2017, 15 (18%) officially registered conservancies generate no cash income at all, and 45 (54%) are unable to fully cover their operating costs (NACSO, 2018, pp. 56, 65). The financial incentives in conservancies may further decrease if donor and NGO support is reduced over time (Boudreaux & Nelson, 2011). Also, rising wildlife numbers are expected to go hand-in-hand with an increase in human-wildlife conflicts (NACSO, 2018, p. 45).

There have been claims in the literature that conservancies can alter the way people relate to the environment, that they buy into the conservation narrative and start protecting nature out of self-interest (Blaikie, 2006). While this is easily imaginable in scenarios where biodiversity conservation can generate income through e.g., ecotourism, it is worthwhile to investigate whether a deeper change of underlying conservation preferences happens in response to CBRNM policies. As stated earlier, using attitudinal items comes with the risk of demand effects and these items are often not validated but constructed ad hoc. Asking for behavioral intentions is problematic as there are clear laws against certain anti-environmental activities like e.g., fire clearing, or illegal hunting practices. Also, people may be reluctant to talk truthfully about certain activities e.g., those they undertake to prevent their crops or even homes from being damaged by elephants. Thus, one may not get to the underlying conservation preference either. Our preferred measure thus is one that asks people to weigh the costs and benefits of conservation which is unrelated to their local conservancy goals and predominantly generates indirect and negligible ecosystem benefits.

3.2. Design of the contingent valuation survey

We designed the survey instrument with the main purpose of eliciting two different measures for conservation preferences. In particular,

Fable 2	
---------	--

Methods for Eliciting Conservation Preferences.

Method	Attitudes	Stated Preferences		Revealed Preferences		
		Self-reported behavior and behavioral intentions	WTP	Observed Behavior	Donations	
Advantages	 Fast and easy to collect Potentially applicable in different contexts (e.g., New Environmental Paradigm Scale, but not always given) Previously tested (reliability/ validity) 	 Change in human behavior is often the targeted outcome of interventions Relatively easy and fast to collect 	- No financial constraints since WTP is hypothetical - Highly flexible and adjustable to specific conservation contexts	- 'First best' solution to measure actual behavioral changes	 Decisions have direct, financial consequences (trade-off) Donation can be chosen to explicitly rule out any direct, economic benefits 	
Disadvantages	- Demand effects - Link to actual conservation behavior not always given	 Interventions may change (dis) incentives to engage in a specific behavior (monetary payoffs, social norms, level of env. quality) Recall bias for self-reported behavior Demand effects Difficult to disentangle motivational drivers (conservation preferences and economic benefits) 	 Hypothetical Bias Demand effect Hypothetical valuation might be unfamiliar (could be reduced with group deliberation and sufficient time) 	 Difficult to disentangle motivational drivers (conservation preferences and economic benefits) Difficult & costly to monitor Lack of experimental control 	 Limited resources reduce the scope to donate even though conservation preferences exist Costly implementation 	

we elicited a) the WTP through a hypothetical donation scenario, and b) real donations through including an opportunity to donate at the end of the survey. The additional inclusion of two different deliberation methods (individual vs group-based) as experimental treatments allows us to assess whether potential differences in conservation preferences between conservancy and non-conservancy villages are sensitive to this design dimension. The sequence of our data collection is illustrated in Fig. 1. A more comprehensive description can be found in the SOM, Section 2.

First, we employed a contingent valuation scenario for a one-time donation to a Namibian conservation project, concerned with monitoring and saving the remaining populations of the lappet-faced vulture (Torgos Tracheliotos), an endangered bird species native to almost all of Namibia. Exploiting the flexibility of the contingent valuation method, we chose to focus our scenario on a species that is not directly related to the income of village inhabitants in this area by any means (as opposed to e.g., a scenario featuring any of the "Big Five" animals attractive for eco-tourism or other aspects related to residents' livelihoods, such as wildfire protection). This design choice directly relates to our argument regarding behavioral changes driven by altered cost/benefit ratios due to conservation interventions. We thereby eliminate potential issues relating to WTP statements being affected by economic self-interest outside of personal budget constraints. The elicited WTP statements (and donations) should be associated mainly with the non-use values (in particular, ecosystem function and existence values) of the lappet-faced vultures. Furthermore, the WTP measure and the actual donations should focus on a species that is known to respondents within and outside of conservancies, but they should not already contribute through other means to the conservation of the specific species. Given the low awareness and knowledge about the lappet-faced vulture among respondents, it is unlikely that people living in the surveyed conservancies might feel that they already contribute to vulture conservation by means other than donating to the project. Neither of the conservancies has explicitly implemented measures to protect vulture populations or any other endangered species. Targeting a species that is more common and directly associated with CBNRM could have influenced the valuation, as conservancy members may already consider their individual contribution to the protection of the species while deciding on their WTP bids and/or donations.

After respondents were provided with basic information on the lappet-faced vulture, they were asked to answer a set of Likert-scale

questions regarding their attitude toward the animal, as well as their personal experience with it. Although no one reported ever seeing a lappet-faced vulture, all respondents indicated that they had seen other kinds of vultures. Thus, in line with our reasoning above, respondents could be expected to value the vultures only for their existence and ecosystem service provision as a contributor to natural pest control (Ogada et al., 2012). We then presented detailed information on the vulture conservation project to the respondents. Immediately before we asked for the WTP amount that respondents hypothetically would be willing to donate to the conservation project, we confronted them with the following cheap-talk script which, as already discussed in section 2.3., have been shown to be an effective tool in mitigating the issue of hypothetical bias in stated preference studies (see e.g., Foster and Burrows, 2017 or Penn and Hu, 2019):

"Before you answer, please think about your response carefully. Keep in mind your personal income situation and alternative possibilities to spend your money. Similar studies showed that people tend to overstate their willingness to pay in hypothetical scenarios, because they do not consider properly how much their spending would affect their personal budget. It is important that you answer as honestly and realistically as possible."

One-time donations allowed us to directly compare the magnitude of the WTP with the donations we elicited at the end of the survey. Naturally, a one-time donation is easier to implement in an area where people have barely any means to make recurring payments, e.g., via mobilepayment applications. To facilitate the respondents' decisions, this was done with the help of so-called payment cards.² We tested for the effect of group deliberation on our different measures of conservation preferences by allocating half of the respondents to small groups of three (see Fig. 1) in which they were asked to jointly discuss their hypothetical donation to the vulture conservation project. Participants were allowed to deliberate for a minimum of three to five minutes. After the deliberation, each respondent reported their individual WTP privately. Stated preference studies increasingly apply group settings in which

² Payment cards elicit individual WTP responses with the help of a range of values presented to the respondents in the form of a list. Following recommendations from established literature on this topic (Bateman et al., 2002), this elicitation format was chosen over other alternatives because it provides a credible context for the possible donation amounts, avoids starting point bias and reduces unrealistic outlier responses. After pretesting, we decided to present respondents with values ranging from 0 NAD to 40 NAD.



Fig. 1. Illustration of Study Sequence.

respondents are given the opportunity to discuss the costs and benefits associated with the good for which WTP is elicited (Kenter, 2016; Spash, 2007). Group deliberation may remedy some of the potential problems inherent to hypothetical valuation methods, including the oftenunrealistic assumption of well-defined and informed preferences among respondents (Schaafsma et al., 2018; Spash, 2007). After the group sessions, the survey interviews continued individually until the end of the session.

After finishing the survey with a set of socio-economic questions, respondents received 30 NAD (4.8 US\$ PPP)³ for their participation at the end of the session and were given the opportunity to donate to the previously presented vulture conservation project (identical to the one in the hypothetical WTP scenario). Respondents donated anonymously by putting money in an envelope and dropping it in a separate box, out of sight of the remaining survey participants and the enumerators. The respective study participant and donation amount could later be linked by an interview ID that was written on the envelope.

The comparison of WTP bids and actual donations allows us to approximate the potential hypothetical bias in WTP bids. As indicated previously, the hypothetical bias states that individuals tend to overstate their WTP in hypothetical scenarios (Carson, 2012b), e.g., due to demand effects (Tourangeau et al., 2000). Such demand effects could be argued to arise both within and outside conservancy villages. In conservancies, respondents might have seen an opportunity to provide a rationale for any kind of government funding (either of monetary nature or via in-kind benefits) in case they indicate positive conservation outcomes by responding accordingly to respective survey items. Outside, villagers on the other hand might have had an interest to respond positively to these survey items to support potential future considerations of their village to become part of a conservancy, as CBNRM conservancies and their alleged benefits are well-known throughout the study area. To mitigate the impact of these demand effects on our conservation preference measures, we clearly stated to not be associated with any kind of governmental agency or NGO that could provide funding and/or help in the establishment of new conservancy areas.

3.3. Study sites for our comparative case study

Our research was concentrated on two conservancies located in northeastern Namibia: The Joseph Mbambangandu Conservancy (JMC), established in 2004, and the George Mukoya Conservancy (GMC), established in 2005. These conservancies were chosen due to being located within the study area of a larger research project but are not atypical when compared to other Namibian case studies. Based on the available data, the GMC and especially the JMC belong to the lowerachieving categories of conservancies in terms of financial revenues (see SOM, Section 1.2). We consider both conservancies prime examples for showcasing the importance of conservation preferences as positive conservation outcomes will likely rely on these underlying drivers due to the absence of tangible financial incentives (corresponding to case C or D in Table 1).

Sampling was conducted in two stages. First, we sampled all four settlements located in the JMC, along with two of the eight villages located in the more remote, barely accessible, and less densely populated GMC⁴. In order to identify "control villages" outside of the conservancies, 58 villages in relative proximity to the main road and the two conservancies were identified using official records. Five localities situated close to the JMC, and two localities situated near the GMC were then randomly selected for the final sample. Twelve participants per village were randomly recruited during village meetings, which were organized by the respective local headman, resulting in a total sample size of 156 observations. Following local ethical research standards, the field team visited all villages a few days before the actual survey work. This was done to acquire approval from the respective village headman, thereby providing them with time to inform all village members about the opportunity to participate and the random nature of the sampling process. All respondents provided informed consent before entering the survey sessions, and they were filled in on data confidentiality and anonymization. Each respondent was remunerated with a fixed participation fee of 30 NAD, which roughly translated to the daily wage for regional farm workers at the time of the study. All interviews took place

 $^{^{3}}$ At the time of the data collection, the monthly minimum wage for farm workers was 900 NAD.

⁴ Of the eight villages in the GMC, two were not accessible by the research team. The two sampled villages were randomly selected from the remaining six villages.

in September 2017 and were held in the local RuKwangali language with the help of three trained research assistants. The socio-economic characteristics of participants by conservancy status are provided in the Supplementary Online Material (SOM Table S5).

3.4. Hypotheses and analytical approach

In line with the performances of the two surveyed conservancies, we expect potentially two mechanisms to affect conservation preferences. On the one hand, the establishment of conservancies may have strengthened overall conservation preferences through training and awareness campaigns as well as deliberation within villages about the value of nature. On the other hand, the fact that the promised benefits of conservancies did not materialize may lead to frustration, potentially reducing conservation preferences. We thus formulate an open hypothesis, acknowledging the two directions of effects:

H1: The average conservation preferences of residents residing within conservancies differ from those of residents living outside of conservancies.

Collecting data on the stated, hypothetical WTP and eliciting actual donations, allows us to approximate the hypothetical bias at the individual level. If we are able to detect systematic differences in the hypothetical bias between study sites, it could be argued that the hypothetical WTP statements cannot provide reliable approximations of relative changes in conservation preferences on their own. Differences in hypothetical bias between study sites could arise from potential demand effects. E.g., within CBNRM conservancies residents may be more susceptible to a social desirability bias. They may feel obliged to answer to the hypothetical questions in line with a positive impact of conservancies and/or may be motivated by reciprocal motives (stating a high valuation of nature as they received funding for the establishment of conservancies by donor agencies and NGOs). This difference in hypothetical WTP would then result in a larger hypothetical bias. Second, residents within conservancies could exhibit more well-defined preferences for ecosystem services. This could be a result of educational interventions and/or collective discussions regarding the significance of conservation activities, both in terms of use and non-use values. Respondents from conservancies could therefore better estimate their true WTP in a hypothetical task. We therefore formulate the following hypotheses:

H2: The hypothetical bias of residents within nature conservancies differs from those living outside of conservancies.

Lastly, we randomly assigned half of our sampled respondents to a treatment condition in which they could discuss their statements with a randomly selected group of fellow community members. Following the argument brought forward in (Spash, 2007), stated preference techniques are under scrutiny for assuming well-informed preferences among respondents, and deliberative monetary valuation (DMV) is presented as a potential remedy to this issue. In the context of our case study, the average effect of group deliberation on the conservation preference measures may differ between study sites. After years of being directly exposed to pro-conservation education and incentivization, people living in conservancies might have more well-defined conservation preferences when compared to people living outside conservancies. If the establishment of conservancies led to more well-defined preferences, group deliberation has potentially a smaller effect on stated WTP and donations in conservancies. Thus, while we expect group deliberation to have a positive effect on both WTP and donations in both village types, we expect these effects to be stronger in magnitude outside of conservancies.

H3: Group deliberation has a positive effect on respondents' WTP and donations, but one that is stronger outside than within conservancies.

We report both non-parametric tests (Mann-Whitney-U tests for continuous outcomes, chi-squared tests for categorical outcomes) and multivariate regressions as hypotheses tests. For Hypotheses 1 we focus on two related, yet unique outcomes: a) the binary decision whether to state a WTP greater than 0 or donate anything, tested with a Probit model (Eq. (1), and b) the WTP and donation amount as a continuous

measure of preferences, tested with a Tobit model (Eq. (2).⁵

The probability of respondent *i* stating a positive WTP or donation is expressed as a function of various explanatory variables, where $\Phi(\cdot)$ denotes the cumulative standard normal distribution:

$$P_i(Y_i > 0|X_i) = \Phi(\beta_0 + Conservancy_i\beta_1 + X_i\beta)$$
⁽¹⁾

where *Conservancy*_i is a dummy variable (taking 1 if the respondent *i* lives in a conservancy and 0 otherwise), and X_i is a vector of control variables including gender, age, age², number of children, household size, education (years of schooling), monthly expenses, and deliberation method (individual vs. group). We report the marginal effects of the Probit model that can be interpreted as the change in the probability of Y > 0 due to a one-unit increase in the explanatory variable. Standard errors are clustered for all regression models at the session level.

As a continuous outcome measure, we use the WTP and donation amount of respondent *i* respectively (y_i) and estimate Tobit models with 0 as lower bound and 40 (30 for the donation) as upper bound to account for the censoring of the data. The Tobit model can be expressed as a latent variable regression model with upper limit *a*, and lower limit *b* (Cong, 2001). The latent continuous variable y_i^* cannot be observed over its entire range, but only within its upper and lower limit:

$$y_i^* = Conservancy_i\beta_1 + X_i\beta + \varepsilon_i \tag{2}$$

$$\mathbf{y}_{i}^{*} = \begin{cases} y_{i}ifb < y_{i} < a \\ bify_{i} \le b \\ aify_{i} \ge a \end{cases}$$
(3)

where *Conservancy*_i is a dummy variable (taking 1 if the respondent *i* lives in a conservancy and 0 otherwise), and X_i is a vector of control variables including gender, age, age², number of children, household size, education (years of schooling), monthly expenses, and deliberation method (individual vs. group). We report the marginal effects that estimate the change in the mean of the latent dependent variable y_i^* due to a one-unit increase in the explanatory variable.

To test Hypothesis 2 we focus on two different measures for the hypothetical bias at the respondent level, namely a) the difference between the stated WTP and donation $(y_i^{bias}, \sec \text{Eq. }(4), \tan b)$ the absolute difference between the stated WTP and donation $(y^{abs-bias}, \sec \text{Eq. }(5))$. We estimate the corresponding Tobit models with these two different dependent variables censored at the respective limits (-30 and 40; 0 and 40).

$$y_i^{bias} = WTP_i - donation_i \tag{4}$$

$$y^{abs-bias}_{i} = |WTP_i - donation_i|$$
(5)

Lastly, we test Hypothesis 3 by estimating the following Tobit regression model for WTP and donations:

$$y_i^* = Conservancy_i\beta_1 + Group_i\beta_2 + Conservancy_i^*Group_i\beta_3 + X_i\beta + \varepsilon_i$$
(6)

where *Conservancy*_i is a dummy variable (taking 1 if respondent *i* lives in a conservancy and 0 otherwise), and *Group*_i is a dummy variable (taking 1 if respondent *i* participated in group deliberation and 0 otherwise). The interaction *Conservancy*_i**Group*_i tests, Hypothesis 3, whether group deliberation has a different effect within conservancies relative to outside. X_i is a vector of control variables including gender, age, age², number of children, household size, education (years of schooling), and monthly expenses.

⁵ Both of these model classes have a rich history of application in conservation research, especially in studies dealing with WTP estimates. Tobit models account for the censored nature of WTP data (Halstead et al., 1991; Norris and Batie, 1987), while Probit models have previously been used to investigate zero and/or protest responses to WTP elicitation questions (Cho et al., 2008).

3.5. Results

3.5.1 WTP and donations

Fig. 2 illustrates the distribution of WTP and donations by conservancy status (see Fig. 2, Panel A and B) as well as the corresponding averages (Fig. 2, Panel D and E; Hypothesis 1). The average WTP in non-conservancies and conservancies are almost identical (Mean = 4.81 vs. 4.75 NAD, SD = 7.55 vs 7.96 NAD, Mann-Whitney-U (MWU) test p = 0.23). However, the average donation from respondents living outside of conservancies is larger (Mean = 3.16 vs. 2.39 NAD, SD = 4.38 vs. 3.55 NAD, MWU p = 0.08). Next, we focus on the categorical outcome whether respondents stated a WTP larger zero and donated anything. Respondents in conservancies are more likely to state a WTP of zero (31%) and donate nothing (44%) compared to the control villages (11% and 26%, respectively). These discrepancies are significantly different, as indicated by chi-squared tests (WTP: p = 0.00, donation: p = 0.02).

In Table 3, we present Probit and Tobit regression models where we control for additional socio-economic covariates (gender, age, house-hold size, education, and monthly expenses). Here, we find that while respondents in conservancies are less likely to state a positive WTP and donate something, average WTP and donations are not significantly different between conservancy and non-conservancy villages.

Finding 1: There is evidence for lower conservation preferences within conservancies based on the hypothetical WTP and actual donation decisions. We observe larger shares of zero donations and zero WTP in conservancies, thus, partially confirming Hypotheses 1.

3.5.2 Hypothetical bias

We can compute the hypothetical bias at the individual level by subtracting the donations from the WTP measure. A positive hypothetical bias thus indicates that respondents overstated their hypothetical WTP in relation to their actual decision. In our sample, 98 out of 156 (63%) respondents show no hypothetical bias. Overall, 45 respondents (29%) have a positive hypothetical bias and 13 (8%) respondents have a negative bias. See Fig. 2, Panel C for the distribution of this variable by conservancy status and Fig. 2, Panel F for the mean hypothetical bias by conservancy status.

On average however, actual donations are substantially lower than the stated WTP (2.80 vs. 4.78 NAD), and the share of respondents who donated nothing is larger than the share of respondents who stated a zero WTP (34.6% vs. 19.9%). The difference between average WTP and donations is statistically significant (Wilcoxon signed-rank test, p < 0.01).

Using WTP as a measure for conservation preferences could be particularly contested if the hypothetical bias would be more or less pronounced in conservancy villages (Hypothesis 2). Here, this difference is not statistically significant (Mean = 1.66 vs 2.36 NAD, SD = 8.42 vs 8.07 NAD, MWU p = 0.75). These findings are also confirmed by Tobit regressions, including socio-economic controls and taking the absolute hypothetical bias as an alternative dependent variable (see Table 4).

Finding 2: There is a significant hypothetical bias, which does not differ significantly between conservancy and non-conservancy villages. We thus do not find support for Hypotheses 2.

3.5.3 Group deliberation

Next, we assess the impact of group deliberation on outcomes. A zero WTP is slightly more common under group than individual deliberation (23 vs 16.7%), even though this difference is not statistically significant (chi-squared test, p = 0.316). The share of respondents who do not donate anything is similar across treatments (group: 35.9%, individual: 33.3%, chi-squared test p = 0.736). We find that prior group deliberation has a negative effect on the average stated individual WTP and donation. The stated WTP is on average 2 NAD lower for the group deliberation treatment (3.76 vs. 5.8 NAD), and donations are on average 1.27 NAD lower for group compared to individual deliberation (2.17 vs.

3.44 NAD). However, these two differences are only statistically significant at the 0.1 level (MWU, WTP: p = 0.09, donation: p = 0.10) based on non-parametric tests and insignificant in regression analysis (see Table 3).⁶ The hypothetical bias (measured as the difference between WTP and donations) is slightly lower under group deliberation (1.60 vs. 2.36 NAD), but this difference is not statistically significant (MWU, p = 0.52).

Through Tobit regressions, we also assess whether group deliberation has different effects in- and outside of conservancies (Hypotheses 3). In the respective models (see Table 3), we include an interaction between the group deliberation and conservancy variable that measures the additional impact of group deliberation in conservancies relative to the effect in non-conservancies. Despite a tendency that group deliberation lowers both WTP and donations in conservancies more than outside of conservancies, these effects are not significant.

Finding 3: We observe a tendency for lower conservation preferences if participants can deliberate in groups before the elicitation of WTP and donations. We do not find evidence that group deliberation has a different effect within and outside of conservancies (Hypotheses 3).

4. Discussion

This article conceptualizes conservation preferences as the drivers of individual decision-making and behavior (Falk et al., 2018; Hausman, 2011) that render people supportive of conservation interventions and also to act in favor of conservation outcomes (Selinske et al., 2020). While they may reflect a wide array of values that people attribute to the conservation of nature (Pascual et al 2023), we argue that they should be distinguishable from preferences for personal economic benefits people may expect to receive from conservation. The article describes and tests hypothetical WTP and real donations within a contingent valuation scenario as a way to measure relative changes in conservation preferences in the context of conservation projects. The major advantage of these two measures is that they are insensitive to demand effects or the altered cost/benefit ratio affecting conservation behavior that is inherent to such projects (Silva & Mosimane, 2014). Our case study in Namibia provides evidence that respondents in conservancies are both less likely to state a positive WTP and less likely to donate any money to the conservation cause. Based on the arguments presented throughout this article, these findings suggest lower average conservation preferences in CBRNM conservancies or at least no increase in conservation preferences due to the existence of the conservancy. We have already alluded to a potential explanation for this finding in the introduction to this paper: it might well be that the conservancy members' average conservation preferences have been undermined by the fact that initially promised economic benefits to the respective communities have failed to materialize over a timeframe of more than ten years (Hegwood et al., 2022; McShane et al., 2011; Muradian et al., 2013).

The comparison between hypothetical WTP and real donations further shows that the hypothetical bias in conservancies is not different from the average bias we measured outside conservancies. From a practical perspective, this finding suggests that WTP elicitation within hypothetical scenarios (i.e., without additional real donation tasks) might be sufficient to assess relative differences in conservation preferences between members and non-members of community-based

⁶ Group and individual deliberation treatments were randomly assigned at the individual level. Respondents in the two treatments had similar observable socio-economic characteristics (see SOM Table S6 for details).



Fig. 2. A) Boxplot of WTP by conservancy status (outliers not displayed), B) Boxplot of donations by conservancy status (outliers not displayed) C) Boxplot of hypothetical bias by conservancy status (WTP_i – *donation*_i), D) Mean WTP by conservancy status (with 95% CIs), E) Mean donation by conservancy status (with 95% CIs), F) Mean hypothetical bias (WTP_i – *donation*_i) by conservancy status (with 95% CIs).

Table 3

Probit (DV: binary outcome positive WTP/donation yes/no; reported as marginal effects) and Tobit model (DV: donation amount) results.

	WTP			Donation		
	(1) Positive WTP (y/n)	(2) Amount (NAD)	(3) Amount (NAD)	(4) Positive Donation (y/n)	(5) Amount (NAD)	(6) Amount (NAD)
Conservancy	-0.21^{**} [-0.34,-0.07]	-1.73 [-4.44,0.98]	0.97 [-3.41,5.36]	-0.21* [-0.39,-0.02]	-1.49 [-3.56,0.59]	-1.04 [-3.93,1.85]
Group Deliberation	-0.05 [-0.21,0.10]	-2.48 [-6.13,1.17]	-0.11 [-4.50,4.28]	-0.02 [-0.14,0.10]	-1.63 [-3.42,0.16]	-1.25 [-3.47,0.98]
Conservancy \times Group Deliberation			-5.45 [-12.99,2.09]			-0.90 [-4.98,3.19]
Obs.	155	155	155	155	155	155
Left-censored		31	31		54	54
Right-censored		5	5		0	0
F-Stat		1.75	1.68		2.80	3.82
Chi2	42.41			17.90		
p-Value	0.00	0.08	0.09	0.04	0.00	0.00
Pseudo R2	0.20	0.02	0.02	0.07	0.03	0.03

Robust standard errors clustered at the session level; *** p < 0.01, **p < 0.05, *p < 0.1; controls: gender, age, age², number of children, household size, education (years of schooling), monthly expenses; Model 1 and 4 report the marginal effects of Probit regressions, while the other models are Tobit regressions. Full models reported in the SOM Tables S1, S2 and S4.

conservation projects. The fact that we observe a positive hypothetical bias in line with the broader contingent valuation literature, suggests that the donation task probably produces the more reliable measures and that attitudinal items are similarly prone to demand effects.⁷

Our results regarding the effects of deliberation on WTP and donation estimates are mixed, and further research is warranted to come to robust recommendations. While we observe a tendency that group deliberation leads to both lower average WTP bids and donations, the effect goes in the opposite direction as we have initially anticipated. While empirical studies have shown that groups can be more selfish relative to individuals (Vollstädt & Böhm, 2019), we additionally look at another circumstance to provide a potential explanation for this finding: Group deliberation does not seem to reduce the hypothetical bias between stated WTP and the real donations. It could thus be that individuals both inside and outside of conservancies may indeed struggle to form and express ad hoc preferences for unfamiliar goods, which is argued to be potentially remedied by group deliberation (Spash, 2007). Our results regarding the slightly lower WTP bids and donations following group deliberation might be a result of the realization that other group members in fact have relatively weak conservation

⁷ Meta-studies on hypothetical bias report that individuals overstate their WTP by a factor of 3 on average, although differences between studies are substantial depending on the type of good studied and the elicitation method applied (List and Gallet, 2001; Murphy et al., 2005; Schläpfer and Fischhoff, 2012). We find that respondents overstate their WTP by a factor of 1.71.

Table 4

To bit regression results with the hypothetical bias $(WTP_i - donation_i)$ as dependent variable.

	Hypothetical Bias (1)	Absolute Hypothetical Bias (2)
Conservancy	0.459	-0.317
	[-1.592,2.510]	[-5.5/1,4.938]
Obs.	155	155
Left-censored	0	98
Right-censored	5	5
F-Stat	2.961	1.235
p-Val	0.003	0.278
Pseudo R2	0.009	0.019

Robust standard errors clustered at the session level; ***p < 0.01, **p < 0.05, *p < 0.1; controls: gender, age, age², number of children, household size, education (years of schooling), monthly expenses; Full models reported in the SOM Table S3.

preferences. If deliberation would be effective in mitigating demand effects, we would likely have observed a larger hypothetical bias within the group deliberation treatment.

Naturally, all of these findings assume that our approach to measuring conservation preferences can be regarded as internally valid in the first place. At first glance, our finding that the measurements of individual monthly expenditures are neither correlated with WTP nor donations may indicate a lack of internal validity. For two reasons, we believe however that this must not necessarily be the case: First, the expenditure measurement might not be a valid indicator of internal validity Gathering income estimates in developing countries is known to be challenging, as households are typically characterized by varying income levels from multiple sources. Because of that, employing consumption expenses instead of income has been considered more reliable (Haughton & Khandker, 2009). Given the limited timeframe to elicit socioeconomic characteristics of respondents in our survey, we relied on a crude measure of expenses through a straightforward mode of questioning. We thus believe that our measure includes additional statistical noise compared to a more detailed approach of eliciting expenses for different categories. Second, even though we observe some degree of variation for the expense variable, the range of observed values remains relatively narrow, likely due to the relatively homogenous nature of our sample. The lack of correlation might therefore be just a result of inelastic demand at the observed variation of income in our sample (Greenstone & Jack, 2015). We would also like to highlight that we collected data on general attitudes towards nature and nature conservation. To fulfill internal validity criteria, one would expect WTP and donations to be correlated with such measures. We do not report average responses to these attitudinal statements in the main text of this paper as three out of these four statements show very little variation (see SOM Table S7), potentially because self-reported agreement with statements is prone to demand effects. To our knowledge, there is no validated and widely accepted attitudinal scale for attitudes towards nature for our study region, which poses a challenge for studies that aim to elicit such attitudes as covariates and do not have the resources to develop and validate contextually adapted measures.

Another reason to treat the results of our case study with at least some degree of caution is the relatively small sample size and hence limited statistical power. Drawing robust conclusions on design considerations (WTP vs. donation; group vs. individual deliberation), would ideally be based on a larger sample and multi-site studies covering different contextual variations. Apart from that, one would ideally have applied a random assignment of the conservation intervention or employed longitudinal data that would allow the application of more robust (quasi)-experimental methods (Ferraro, 2009; Greenstone & Gayer, 2009). Nonetheless, we find a few reasons to assume pre-existing differences between villages with respect to our outcome variables: First, the control villages are located close to the conservancy villages. On average, the control villages are located 18.4 km away from the nearest conservancy village (min. 6.3 km; max. 36.2 km). As a result, villages inside and outside of conservancies likely share similar socioeconomic and ecological conditions and thus have equal potential for CBNRM success and economic development in general. Second, both our surveyed conservancies were initiated by community members rather than external stakeholders, such as NGOs or the Namibian government, who could have targeted certain communities selectively. To the best of our knowledge, there have been no (failed) attempts to establish conservancies in any of our control villages. Thus, there might if at all be a positive selection bias stemming from the initial group of conservancy founders. Assuming the non-random establishment of conservancies (i. e., if conservation preferences were initially higher in the pre-CBNRM conservancy villages), the true conservancy effect would be biased upwards and thus be lower than what we observe based on our data.

On the subject of the external validity of our case study, we would again like to point out that we did purposefully choose the two conservancies because they do not report noteworthy financial revenues through e.g., ecotourism, so we assume the formation of conservation preferences especially important (see cases C and D in Table 1). Against this backdrop, we could only reasonably generalize our findings for conservancies with comparable financial performance. Many of the Namibian conservancies, especially in the Kunene and Erongo regions of northwestern Namibia, report significantly higher financial revenue streams (NACSO, 2018). Within the subgroup of conservancies comparable to the ones we have sampled, we do however believe that are results can be regarded as externally valid, as we have implemented a random sampling mechanism. We have also identified two studies in which researchers have elicited a one-time WTP for the conservation of threatened vulture species using contingent valuation, which we therefore regard at least somewhat comparable to our approach. While one Israeli study using payment cards has produced comparable average WTP bids for vulture conservation when adjusting WTP bids for inflation and national per capita income (Becker et al., 2009), another study reports significantly higher average WTP bids (Zambrano-Monserrate, 2020). However, this study elicited the WTP via dichotomous choice with randomly allocated bids including relatively high monetary figures when compared to our study. WTP elicitation methods of this type are also known to yield higher WTP bids than payment card approaches (Blaine et al., 2005).

We can also not rule out that our WTP estimates might have been affected by the participation fee of 30 NAD which was, of course, known to the respondents prior to the start of the individual surveys. Naturally, this circumstance could have the potential to negatively affect the external validity of the participants' stated WTP bids. However, such effects should be leveled between respondents from both conservancy and non-conservancy villages. Thus, with regards to our study's main research objective, which was to investigate mean WTP differences between village types, we do not regard this as a major cause for concern. However, we did try to mitigate such effects in different ways: We addressed potential WTP anchoring effects caused by the participation fee by offering WTP bids higher than the fee on our payment cards. The observed distribution of our WTP estimates towards the lower end of the payment card spectrum does indicate that anchoring towards the 30 NAD value did not seem to be a prevalent problem. Also, respondents were made aware of the payment a few days prior to the study (during the recruitment phase). This time horizon likely reduced the "windfall gain" effect compared to a "surprise" earning at the time of the interview, as participants had a longer time to internalize the money and make plans for its use (Arkes et al., 1994). Still, future studies could further try to circumvent problems of this nature by adjusting the survey sequence, tweaking the elicitation method of the contingent valuation method (e.g., open-ended, bidding game, or dichotomous choice), or the mode of reimbursement to the study participants.

5. Conclusion

In this article, we showcased how the combination of a contingent valuation scenario and a real donation task can be applied to measure and investigate conservation preferences. We argued that knowledge about such preferences can help conservation policymakers better evaluate the long-term impacts of conservation projects, which is especially crucial in cases where financial incentives to affected communities do not materialize as initially promised. Our proposed approach can serve as a complement to other preference measurement instruments in conservation contexts, as it avoids various pitfalls associated with other methodological approaches, which have been highlighted in section 2 of the paper at hand. The outcomes of the case study surrounding the Namibian CBNRM program, in which we have tested the proposed measurement approach, suggest that conservation preferences can indeed fail to materialize or even be undermined.

We argue that a careful design of the (hypothetical) donation cause is of crucial importance. Following our arguments brought forward specifically in Sections 1 and 2 of this article, future studies should aim to further minimize the potential role of direct use values (in the sense of a direct economic benefit of the donation cause to the respondents) in contingent valuation scenarios aiming to elicit conservation preferences. It should furthermore be ensured that the species or ecosystem central to the valuation scenario is comparably familiar to respondents in the relevant sites but not subject to conservation efforts already. We ultimately hope that the conceptualization of conservation preferences and the methodological contribution brought forward in this article can inform and inspire future studies that want to measure conservation preferences as an outcome of policy intervention.

CRediT authorship contribution statement

Marco Nilgen: Writing – review & editing, Writing – original draft, Project administration, Methodology, Formal analysis, Data curation, Conceptualization. **Julian Rode:** Writing – review & editing, Writing – original draft, Validation, Supervision, Methodology, Conceptualization. **Tobias Vorlaufer:** Writing – review & editing, Writing – original draft, Methodology, Investigation, Formal analysis, Data curation, Conceptualization. **Björn Vollan:** Writing – review & editing, Writing – original draft, Supervision, Project administration, Methodology, Investigation, Funding acquisition, Conceptualization.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

Data will be made available on request.

Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.ecoser.2024.101621.

References

- Achen, C. H., 1975. Mass political attitudes and the survey response. American Political Science Review 69 (4), 1218–1231. doi: https://doi.org/10.2307/1955282.
- Anderson, C.B., Athayde, S., Raymond, C.M., Vatn, A., Arias, P., Gould, R.K., Kenter, J., Muraca, B., Sachdeva, S., Samakov, A., Zent, E., Lenzi, D., Murali, R., Amin, A., Cantú-Fernández, M., 2022. Conceptualizing the diverse values of nature and their contributions to people. In: Methodological Assessment Report on the Diverse Values and Valuation of Nature of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services. IPBES secretariat. https://doi.org/10.5281/ zenodo.6493134.

- Andor, M.A., Frondel, M., Vance, C., 2017. Mitigating hypothetical bias: evidence on the effects of correctives from a large field study. Environ. Resour. Econ. 68 (3), 777–796. https://doi.org/10.1007/S10640-016-0047-X/TABLES/7.
- Arkes, H.R., Joyner, C.A., Pezzo, M.V., Nash, J.G., Siegel-Jacobs, K., Stone, E., 1994. The psychology of windfall gains. Organ. Behav. Hum. Decis. Process. 59 (3), 331–347. https://doi.org/10.1006/obhd.1994.1063.
- Armitage, C.J., Conner, M., 2001. Efficacy of the theory of planned behaviour: a metaanalytic review. Br. J. Soc. Psychol. 40 (4), 471–499. https://doi.org/10.1348/ 014466601164939.
- Bastien-Olvera, B.A., Moore, F.C., 2021. Use and non-use value of nature and the social cost of carbon. Nat. Sustainability 4, 101–108. https://doi.org/10.1038/s41893-020-00615-0.
- Bateman, I.J., Carson, R.T., Day, B., Hanemann, M., Hanley, N., Hett, T., Jones-Lee, M., Loomes, G., 2002. Economic Valuation with Stated Preference Techniques: A Manual. Edward Elgar Publishing doi: 10.4337/9781781009727.
- Becker, N., Choresh, Y., Bahat, O., Inbar, M., 2009. Economic analysis of feeding stations as a means to preserve an endangered species: the case of Griffon Vulture (Gyps fulvus) in Israel. J. Nat. Conserv. 17 (4), 199–211. https://doi.org/10.1016/J. JNC.2009.04.004.
- Blaikie, P., 2006. Is small really beautiful? community-based natural resource management in Malawi and Botswana. World Dev. 34 (11), 1942–1957. https://doi. org/10.1016/J.WORLDDEV.2005.11.023.
- Blaine, T.W., Lichtkoppler, F.R., Jones, K.R., Zondag, R.H., 2005. An assessment of household willingness to pay for curbside recycling: a comparison of payment card and referendum approaches. J. Environ. Manage. 76 (1), 15–22. https://doi.org/ 10.1016/J.JENVMAN.2005.01.004.
- Blanco, E., Lopez, M.C., Coleman, E.A., 2012. Voting for environmental donations: experimental evidence from Majorca, Spain. Ecol. Econ. 75, 52–60. https://doi.org/ 10.1016/j.ecolecon.2011.12.014.
- Blanco, E., Moros, L., Pfaff, A., Steimanis, I., Velez, M.A., Vollan, B., 2023. No crowding out among those terminated from an ongoing PES program in Colombia. J. Environ. Econ. Manag. 120, 102826 https://doi.org/10.1016/J.JEEM.2023.102826.
- Boudreaux, K., Nelson, F., 2011. Community conservation in Namibia: empowering the poor with property rights. Econ. Aff. 31 (2), 17–24. https://doi.org/10.1111/j.1468-0270.2011.02096.x.
- Bouma, J.A., Koetse, M.J., 2019. Mind the gap: stated versus revealed donations and the differential role of behavioral factors. Land Econ. 95 (2), 225–245. https://doi.org/ 10.3368/le.95.2.225.
- Bowles, S., 1998. Endogenous preferences: The cultural consequences of markets and other economic institutions. J. Econ. Lit. 36 (1), 75–111.doi. https://www.jstor.org/ stable/2564952.
- Carson, R.T., 2012. Contingent valuation: a practical alternative when prices aren't available. J. Econ. Perspect. 26 (4), 27–42. https://doi.org/10.1257/JEP.26.4.27.
- Carson, R.T., Mitchell, R.C., Hanemann, M., Kopp, R.J., Presser, S., Ruud, P.A., 2003. Contingent valuation and lost passive use: damages from the Exxon Valdez oil spill. Environ. Resour. Econ. 25 (3), 257–286. https://doi.org/10.1023/A: 1024486702104.
- Chaikaew, P., Hodges, A.W., Grunwald, S., 2017. Estimating the value of ecosystem services in a mixed-use watershed: a choice experiment approach. Ecosyst. Serv. 23, 228–237. https://doi.org/10.1016/J.ECOSER.2016.12.015.
- Champ, P.A., Bishop, R.C., Brown, T.C., McCollum, D.W., 1997. Using donation mechanisms to value nonuse benefits from public goods. J. Environ. Econ. Manag. 33 (2), 151–162. https://doi.org/10.1006/jeem.1997.0988.
- Chan, K.M.A., Balvanera, P., Benessaiah, K., Chapman, M., Díaz, S., Gómez-Baggethun, E., Gould, R., Hannahs, N., Jax, K., Klain, S., Luck, G.W., Martín-López, B., Muraca, B., Norton, B., Ott, K., Pascual, U., Satterfield, T., Tadaki, M., Taggart, J., Turner, N., 2016. Why protect nature? Rethinking values and the environment. PNAS 113 (6), 1462–1465. https://doi.org/10.1073/ PNAS.1525002113/ASSET/B03A0DE1-AEDA-4A3C-9FF5-8AD6256F16B4/ASSETS/ PNAS.1525002113.FP.PNG.
- Cho, S.-H., Yen, S.T., Bowker, J.M., Newman, D.H., 2008. Modeling willingness to pay for land conservation easements: treatment of zero and protest bids and application and policy implications. J. Agric. Appl. Econ. 40 (1), 267–285. https://doi.org/ 10.1017/S1074070800028108.
- Cong, R., 2001. Marginal effects of the tobit model. Stata Tech. Bull. 10 (56). https://E conPapers.repec.org/RePEc:tsj:stbull:y:2001:v:10:i:56:sg144.
- de Groot, J.I.M., Steg, L., 2008. Value orientations to explain beliefs related to environmental significant behavior. Environ. Behav. 40 (3), 330–354. https://doi. org/10.1177/0013916506297831.
- Dias, V., Belcher, K., 2015. Value and provision of ecosystem services from prairie wetlands: a choice experiment approach. Ecosyst. Serv. 15, 35–44. https://doi.org/ 10.1016/J.ECOSER.2015.07.004.
- Dunlap, R.E., van Liere, K.D., Mertig, A.G., Jones, R.E., 2000. New trends in measuring environmental attitudes: measuring endorsement of the new ecological paradigm: a revised NEP scale. J. Soc. Issues 56 (3), 425–442. https://doi.org/10.1111/0022-4537.00176.
- Ezzine-de-Blas, D., Corbera, E., Lapeyre, R., 2019. Payments for environmental services and motivation crowding: towards a conceptual framework. Ecol. Econ. 156, 434–443. https://doi.org/10.1016/j.ecolecon.2018.07.026.
- Falk, A., Becker, A., Dohmen, T., Enke, B., Huffman, D., Sunde, U., 2018. Global evidence on economic preferences*. Q. J. Econ. 133 (4), 1645–1692. https://doi.org/ 10.1093/qje/qjy013.
- Farjam, M., Nikolaychuk, O., Bravo, G., 2019. Experimental evidence of an environmental attitude-behavior gap in high-cost situations. Ecol. Econ. 166, 106434 https://doi.org/10.1016/j.ecolecon.2019.106434.

- Farrell, J., Rabin, M., 1996. Cheap talk. J. Econ. Perspect. 10 (3), 103–118. https://doi. org/10.1257/jep.10.3.103.
- Ferraro, P.J., 2009. Counterfactual thinking and impact evaluation in environmental policy. N. Dir. Eval. 2009 (122), 75–84. https://doi.org/10.1002/ev.297.
- Fishbein, M., Ajzen, I., 1975. Belief, Attitude, Intention and Behavior: An Introduction to Theory and Research. Addison-Wesley, Reading, Mass. [u.a.]
- Floress, K., Reimer, A., Thompson, A., Burbach, M., Knutson, C., Prokopy, L., Ribaudo, M., Ulrich-Schad, J., 2018. Measuring farmer conservation behaviors: challenges and best practices. Land Use Policy 70, 414–418. https://doi.org/ 10.1016/j.landusepol.2017.11.030.
- Foster, H., Burrows, J., 2017. Hypothetical bias: a new meta-analysis. In: McFadden, D., Train, K. (Eds.), Contingent Valuation of Environmental Goods: A Comprehensive Critique. Edward Elgar Publishing, pp. 270–291.
- Frey, B.S., Stutzer, A., 1999. Measuring preferences by subjective well-being. J. Inst. Theor. Econ. 155 (4), 755–778. https://www.jstor.org/stable/40752166.
- Gifford, R., Kormos, C., McIntyre, A., 2011. Behavioral dimensions of climate change: drivers, responses, barriers, and interventions. Wiley Interdiscip. Rev. Clim. Change 2 (6), 801–827. doi: https://doi.org/10.1002/wcc.143.
- Green, K.M., Crawford, B.A., Williamson, K.A., DeWan, A.A., 2019. A meta-analysis of social marketing campaigns to improve global conservation outcomes. Soc. Mark. Q. 25 (1), 69–87. https://doi.org/10.1177/1524500418824258.
- Greenstone, M., Gayer, T., 2009. Quasi-experimental and experimental approaches to environmental economics. J. Environ. Econ. Manag. 57 (1), 21–44. https://doi.org/ 10.1016/j.jeem.2008.02.004.
- Greenstone, M., Jack, B.K., 2015. Envirodevonomics: a research agenda for an emerging field. J. Econ. Lit. 53 (1), 5–42. https://doi.org/10.1257/JEL53.1.5.
- Halstead, J.M., Lindsay, B.E., Brown, C.M., 1991. Use of the tobit model in contingent valuation: experimental evidence from the Pemigewasset Wilderness Area. J. Environ. Manage. 33 (1), 79–89. https://doi.org/10.1016/S0301-4797(05)80049-0.
- Haughton, J., Khandker, S.R., 2009. Handbook on Poverty & Inequality. The World Bank.
- Hausman, J., 2012. Contingent valuation: from dubious to hopeless. J. Econ. Perspect. 26 (4), 43–56. https://doi.org/10.1257/JEP.26.4.43.
- Hausman, D.M., 2011. Preference, Value, Choice, and Welfare. Cambridge University Press. doi: 10.1017/CBO9781139058537.
- Hayes, T., Murtinho, F., Wolff, H., López-Sandoval, M.F., Salazar, J., 2022. Effectiveness of payment for ecosystem services after loss and uncertainty of compensation. Nat. Sustainability 5 (1), 81–88. https://doi.org/10.1038/s41893-021-00804-5.
- Hegwood, M., Langendorf, R.E., Burgess, M.G., 2022. Why win-wins are rare in complex environmental management. Nat. Sustainability 5 (8), 674–680. https://doi.org/ 10.1038/s41893-022-00866-z.
- Hensher, D.A., 2010. Hypothetical bias, choice experiments and willingness to pay. Transp. Res. B Methodol. 44 (6), 735–752. https://doi.org/10.1016/J. TRB.2009.12.012.
- Hsee, C.K., Rottenstreich, Y., 2004. Music, pandas, and muggers: on the affective psychology of value. J. Exp. Psychol. Gen. 133 (1), 23–30. https://doi.org/10.1037/ 0096-3445.133.1.23.
- Kenny, J., 2021. Are environment versus economy trade-off questions more about environmental or economic attitudes? Int. J. Public Opin. Res. 33 (1), 159–170. https://doi.org/10.1093/IJPOR/EDAA003.
- Kenter, J.O., 2016. Integrating deliberative monetary valuation, systems modelling and participatory mapping to assess shared values of ecosystem services. Ecosyst. Serv. 21, 291–307. https://doi.org/10.1016/j.ecoser.2016.06.010.
- Kling, C.L., Phaneuf, D.J., Zhao, J., 2012. From Exxon to BP: has some number become better than no number? J. Econ. Perspect. 26 (4), 3–26. https://doi.org/10.1257/ jep.26.4.3.
- Kollmuss, A., Agyeman, J., 2002. Mind the Gap: Why do people act environmentally and what are the barriers to pro-environmental behavior? Environ. Educ. Res. 8 (3), 239–260. https://doi.org/10.1080/13504620220145401.
- Krutilla, J.V., 1967. Conservation reconsidered. Am. Econ. Rev. 57 (4), 777–786. https:// www.jstor.org/stable/1815368.
- List, J.A., Gallet, C.A., 2001. What experimental protocol influence disparities between actual and hypothetical stated values? Environ. Resour. Econ. 20 (3), 241–254. https://doi.org/10.1023/A:1012791822804.
- Lopes, A.F., Kipperberg, G., 2020. Diagnosing insensitivity to scope in contingent valuation. Environ. Resour. Econ. 77 (1), 191–216. https://doi.org/10.1007/ S10640-020-00470-9/TABLES/3.
- Manero, A., Taylor, K., Nikolakis, W., Adamowicz, W., Marshall, V., Spencer-Cotton, A., Nguyen, M., Grafton, R.Q., 2022. A systematic literature review of non-market valuation of Indigenous peoples' values: current knowledge, best-practice and framing questions for future research. Ecosyst. Serv. 54, 101417 https://doi.org/ 10.1016/J.ECOSER.2022.101417.
- Martínez-Alier, J., 2002. The Environmentalism of the Poor A Study of Ecological Conflicts and Valuation. Edward Elgar.
- McShane, T.O., Hirsch, P.D., Trung, T.C., Songorwa, A.N., Kinzig, A., Monteferri, B., Mutekanga, D., van Thang, H., Dammert, J.L., Pulgar-Vidal, M., Welch-Devine, M., Peter Brosius, J., Coppolillo, P., O'Connor, S., 2011. Hard choices: making trade-offs between biodiversity conservation and human well-being. Biol. Conserv. 144 (3), 966–972. https://doi.org/10.1016/j.biocon.2010.04.038.
- Milfont, T.L., 2009. The effects of social desirability on self-reported environmental attitudes and ecological behaviour. Environmentalist 29 (3), 263–269. https://doi. org/10.1007/S10669-008-9192-2/TABLES/1.
- Muradian, R., Arsel, M., Pellegrini, L., Adaman, F., Aguilar, B., Agarwal, B., Corbera, E., Ezzine de Blas, D., Farley, J., Froger, G., Garcia-Frapolli, E., Gómez-Baggethun, E., Gowdy, J., Kosoy, N., Le Coq, J.F., Leroy, P., May, P., Méral, P., Mibielli, P.,

Norgaard, R., Ozkaynak, B., Pascual, U., Pengue, W., Perez, M., Pesche, D., Pirard, R., Ramos-Martin, J., Rival, L., Saenz, F., Van Hecken, G., Vatn, A., Vira, B., Urama, K., 2013. Payments for ecosystem services and the fatal attraction of win-win solutions. Conserv. Lett. 6 (4), 274–279.

- Murphy, J.J., Allen, P.G., Stevens, T.H., Weatherhead, D., 2005. A Meta-analysis of Hypothetical Bias in Stated Preference Valuation. Environ. Resour. Econ. 30 (3), 313–325. https://doi.org/10.1007/s10640-004-3332-z.
- NACSO, 2018. The state of community conservation in Namibia a review of communal conservancies, community forests and other CBNRM initiatives (2017 Annual Report). Namibian Association of CBNRM Support Organisations (NACSO). http:// www.nacso.org.na/sites/default/files/State%20of%20Community%20Conservation %20book%20web_0.pdf.
- Nilsson, D., Fielding, K., Dean, A.J., 2020. Achieving conservation impact by shifting focus from human attitudes to behaviors. Conserv. Biol. 34 (1), 93–102. https://doi. org/10.1111/cobi.13363.
- Norris, P.E., Batie, S.S., 1987. Virginia farmers' soil conservation decisions: an application of tobit analysis. J. Agric. Appl. Econ. 19 (1), 79–90. https://doi.org/ 10.1017/S0081305200017404.
- Ogada, D.L., Torchin, M.E., Kinnaird, M.F., Ezenwa, V.O., 2012. Effects of vulture declines on facultative scavengers and potential implications for mammalian disease transmission. Conserv. Biol. 26 (3), 453–460. https://doi.org/10.1111/j.1523-1739.2012.01827.x.
- Otrachshenko, V., Tyurina, E., Nagapetyan, A., 2022. The economic value of the Glass Beach: contingent valuation and life satisfaction approaches. Ecol. Econ. 198, 107466 https://doi.org/10.1016/J.ECOLECON.2022.107466.
- Pascual, U., Balvanera, P., Díaz, S., Pataki, G., Roth, E., Stenseke, M., Watson, R.T., Başak Dessane, E., Islar, M., Kelemen, E., Maris, V., Quaas, M., Subramanian, S.M., Wittmer, H., Adlan, A., Ahn, SoEun, Al-Hafedh, Y.S., Amankwah, E., Asah, S.T., Berry, P., Bilgin, A., Breslow, S.J., Bullock, C., Cáceres, D., Daly-Hassen, H., Figueroa, E., Golden, C.D., Gómez-Baggethun, E., González-Jiménez, D., Houdet, J., Keune, H., Kumar, R., Ma, K., May, P.H., Mead, A., O'Farrell, P., Pandit, R., Pengue, W., Pichis-Madruga, R., Popa, F., Preston, S., Pacheco-Balanza, D., Saarikoski, H., Strassburg, B.B., van den Belt, M., Verma, M., Wickson, F., Yagi, N., 2017. Valuing nature's contributions to people: the IPBES approach. Curr. Opin. Environ. Sustain. 26-27, 7–16.
- Pascual, U., Balvanera, P., Anderson, C.B., Chaplin-Kramer, R., Christie, M., González-Jiménez, D., Martin, A., Raymond, C.M., Termansen, M., Vatn, A., Athayde, S., Baptiste, B., Barton, D.N., Jacobs, S., Kelemen, E., Kumar, R., Lazos, E., Mwampamba, T.H., Nakangu, B., Zent, E., 2023. Diverse values of nature for sustainability. Nature 620 (7975), 813–823. https://doi.org/10.1038/s41586-023-06406-9.
- Pascual, U., Muradian, R., Brander, L., Martín-López, B., Gómez-Baggethun, E., Verma Contributing, M., Armsworth, P., Christie, M., Cornelissen, H., Eppink, F., Farley, J., Loomis, J., Pearson, L., Perrings, C., Reviewers, S. P., Mcneely, J. A., Norgaard, R., Siddiqui, R., Simpson, R. D., & Turner, R.K., 2010. The Economics of Valuing Ecosystem Services and Biodiversity. In: The Economics of Ecosystems and Biodiversity: The Ecological and Economic Foundations. https://www.researchgate. net/publication/303444184.
- Penn, J.M., Hu, W., 2018. Understanding hypothetical bias: an enhanced meta-analysis. Am. J. Agric. Econ. 100 (4), 1186–1206. https://doi.org/10.1093/AJAE/AAY021.
- Penn, J., Hu, W., 2019. Cheap talk efficacy under potential and actual hypothetical bias: a meta-analysis. J. Environ. Econ. Manag. 96, 22–35. https://doi.org/10.1016/j. jeem.2019.02.005.
- Rode, J., Hogarth, R.M., le Menestrel, M., 2008. Ethical differentiation and market behavior: an experimental approach. J. Econ. Behav. Organ. 66 (2), 265–280. https://doi.org/10.1016/j.jebo.2006.12.003.
- Rode, J., Gómez-Baggethun, E., Krause, T., 2015. Motivation crowding by economic incentives in conservation policy: a review of the empirical evidence. Ecol. Econ. 117, 270–282. https://doi.org/10.1016/j.ecolecon.2014.11.019.
- Ryan, R.M., Deci, E.L., 2000. Self-determination theory and the facilitation of intrinsic motivation, social development, and well-being. Am. Psychol. 55 (1), 68–78. https://doi.org/10.1037/0003-066X.55.1.68.
- Schaafsma, M., Bartkowski, B., Lienhoop, N., 2018. Guidance for deliberative monetary valuation studies. Int. Rev. Environ. Resour. Econ. 12 (2–3), 267–323. https://doi. org/10.1561/101.00000103.
- Schläpfer, F., Fischhoff, B., 2012. Task familiarity and contextual cues predict hypothetical bias in a meta-analysis of stated preference studies. Ecol. Econ. 81, 44–47. https://doi.org/10.1016/j.ecolecon.2012.06.016.
- Selinske, M.J., Garrard, G.E., Gregg, E.A., Kusmanoff, A.M., Kidd, L.R., Cullen, M.T., Cooper, M., Geary, W.L., Hatty, M.A., Hames, F., Kneebone, S., McLeod, E.M., Ritchie, E.G., Squires, Z.E., Thomas, J., Willcock, M.A.W., Blair, S., Bekessy, S.A., 2020. Identifying and prioritizing human behaviors that benefit biodiversity. Conserv. Sci. Pract. 2 (9) https://doi.org/10.1111/csp2.249.
- Shiell, A., Hawe, P., Seymour, J., 1997. Values and preferences are not necessarily the same. Health Econ. 6 (5), 515–518. https://doi.org/10.1002/(SICI)1099-1050 (199709)6:5<515::AID-HEC292>3.0.CO;2-N.
- Silva, J.A., Mosimane, A., 2014. "How could I live here and not be a member?": Economic versus social drivers of participation in Namibian conservation programs. Hum. Ecol. 42 (2), 183–197. https://doi.org/10.1007/S10745-014-9645-9/ TABLES/7.
- Spash, C.L., 2007. Deliberative monetary valuation (DMV): issues in combining economic and political processes to value environmental change. Ecol. Econ. 63 (4), 690–699.
- Spash, C.L., Urama, K., Burton, R., Kenyon, W., Shannon, P., Hill, G., 2009. Motives behind willingness to pay for improving biodiversity in a water ecosystem:

M. Nilgen et al.

economics, ethics and social psychology. Ecol. Econ. 68 (4), 955–964. https://doi. org/10.1016/j.ecolecon.2006.09.013.

- Stern, P.C., Dietz, T., 1994. The value basis of environmental concern. J. Soc. Issues 50 (3), 65–84. https://doi.org/10.1111/j.1540-4560.1994.tb02420.x.
- Stevens, T.H., Tabatabaei, M., Lass, D., 2013. Oaths and hypothetical bias. J. Environ. Manage. 127, 135–141. https://doi.org/10.1016/J.JENVMAN.2013.04.038.
- Tourangeau, R., Rips, L.J., Rasinski, K., 2000. The Psychology of Survey Response. Cambridge University Press doi: 10.1017/CB09780511819322.
- Vesely, S., Klöckner, C.A., 2020. Social desirability in environmental psychology research: three meta-analyses. Front. Psychol. 11, 545057 https://doi.org/10.3389/ FPSYG.2020.01395/BIBTEX.
- Vollstädt, U., Böhm, R., 2019. Are groups more competitive, more selfish-rational or more prosocial bargainers? J. Behav. Exp. Econ. 78, 146–159. https://doi.org/ 10.1016/J.SOCEC.2018.12.003.
- Webb, T.L., Sheeran, P., 2006. Does changing behavioral intentions engender behavior change? A meta-analysis of the experimental evidence. Psychol. Bull. 132 (2), 249–268. https://doi.org/10.1037/0033-2909.132.2.249.
- Weisbrod, B.A., 1964. Collective-consumption services of individual-consumption goods. Q. J. Econ. 78 (3), 471–477. https://doi.org/10.2307/1879478.
- Wuepper, D., Clemm, A., Wree, P., 2019. The preference for sustainable coffee and a new approach for dealing with hypothetical bias. J. Econ. Behav. Organ. 158, 475–486. https://doi.org/10.1016/J.JEBO.2018.12.012.
- Zambrano-Monserrate, M.A., 2020. The economic value of the Andean Condor: the national symbol of South America. J. Nat. Conserv. 54, 125796 https://doi.org/ 10.1016/J.JNC.2020.125796.
- Zizzo, D.J., 2010. Experimenter demand effects in economic experiments. Exp. Econ. 13 (1), 75–98. https://doi.org/10.1007/s10683-009-9230-z.
- von Weizsäcker, C. C. (2005). The welfare economics of adaptive preferences. MPI Collective Goods Preprint, (2005/11). doi: https://doi.org/10.2139/ssrn.771904.